Author's response to reviews

Title: Panobinostat synergizes with bortezomib to induce endoplasmic reticulum stress and ubiquitinated protein accumulation in renal cancer cells

Authors:

Akinori Sato (zenpaku@ndmc.ac.jp)
Takako Asano (urotaka5@ndmc.ac.jp)
Makoto Isono (drur589@ndmc.ac.jp)
Keiichi Ito (itok@ndmc.ac.jp)
Tomohiko Asano (otosan@ndmc.ac.jp)

Version: 2
Date: 22 March 2014

Author's response to reviews:

Dear Reymon Ampalaya
Journal Editorial Office
BioMed Central

With regard to this research project we don't have anyone whom we should acknowledge and any source of funding to be named in the "Acknowledgment section". We added an "Acknowledgment section", but simply wrote "none". Also, we added line numbers to the manuscript as suggested. Thank you very much for your kind assistance and we are looking forward to hearing from you soon.

Sincerely yours,

Akinori Sato MD, PhD
Department of Urology
National Defense Medical College